Author’s response to reviews

Title: Retinal vasculitis with Chronic Recurrent Multifocal Osteomyelitis: a case report and review of the literature

Authors:

Victoria Shanmugam (vshanmugam@mfa.gwu.edu)
Marc Phillipotts (mphillpotts@mfa.gwu.edu)
Timothy Brady (tim_brady@gwmail.gwu.edu)
Monica Dalal (mdalal@mfa.gwu.edu)
Shawn Haji-Momenian (shajimomenian@mfa.gwu.edu)
Esma Akin (eakin@mfa.gwu.edu)
Kavita Nataranjan (knataranjan@mfa.gwu.edu)
Sean McNish (mcnish23@gmail.com)
Donald Karcher (dkarcher@mfa.gwu.edu)

Version: 2 Date: 06 Mar 2019

Author’s response to reviews:

Dear Dr. Gaffo,

RE: Retinal vasculitis with Chronic Recurrent Multifocal Osteomyelitis: a case report and review of the literature (BRHM-D-18-00087)

Thank you for reviewing the attached manuscript for possible publication in the BMC Rheumatology. We have responded to reviewer comments below.

1. Page 6, Table 3 makes an appearance in Line 9 before tables 1 and 2 referenced in Page 8. This surely happened because of the rearrangement of the contents requested by reviewers, but needs to be corrected.

This change has been made as requested.
2. CT abbreviation is defined for the first time in page 6. However makes first appearance in Page 4, line 20.

This change has been made as requested.

3. MRI is also not defined in the same paragraph

This change has been made as requested.

4. Consider removing brand name of adalimumab Page 18, line 14.

This change has been made as requested.